Author’s response to reviews

Title: Inflammatory myoglandular polyp of the cecum: case report and review of literature

Authors:

Roberto L Meniconi (robmeni@tiscali.it)
Roberto Caronna (roberto.caronna@uniroma1.it)
Michele Benedetti (micbenei@tin.it)
Gianfranco Fanello (gianfranco.fanello@uniroma1.it)
Antonio Ciardi (antonio.ciardi@uniroma1.it)
Monica Schiratti (monica.schiratti@uniroma1.it)
Federica Papini (federica.papini@libero.it)
Francesco Farelli (frafa83@gmail.it)
Giuseppe Dinatale (giuseppedinatalef4@tiscali.it)
Piero Chirletti (piero.chirletti@uniroma1.it)

Version: 3 Date: 20 January 2010

Author’s response to reviews: see over
To: BMC Editorial Production Team

MAJOR

1) We revised the References and cited them using consecutive numbers in square brackets, as requested.

MINOR

1) We presented all text in black
2) We made linguistic corrections as requested by reviewers